Author’s response to reviews

Title: Exceptional serological and radiological response to sorafenib in 2 patients with advanced hepatocellular carcinoma and chronic hepatitis C viral infection: Case report and review of the literature.

Authors:

Catherine Atkin (catherine.atkin@uhb.nhs.uk)
Philip Earwaker (philip.earwaker@uhb.nhs.uk)
Arvind Pallan (arvind.pallan@uhb.nhs.uk)
Shishir Shetty (shishir.shetty@uhb.nhs.uk)
Pankaj Punia (pankaj.punia@uhb.nhs.uk)
Yuk Ting Ma (y.t.ma@bham.ac.uk)

Version: 3 Date: 18 Jan 2017

Author’s response to reviews:

Thank you for including the Declarations statements. Because your manuscript does report patient data, the statement “Not applicable” is not appropriate for the Availability of data and material statement, please describe the availability of any data collected or state that all the data are contained in the manuscript. Similarly, if no funding was received, please specifically state this in the Funding statement.

These subsections have now been amended.